\documentclass{article}
\usepackage{fullpage}
\begin{document}
\begin{itemize}
\item[1/23] Optional: \cite{Ernst2003}
\item[1/28] \cite{ChowPGCR2004,NewsomeS2005}
\item[1/30] \cite{SlowinskaB2009,BosmanSB2011}
            Historic: \cite{Wall1989}
            Optional: \cite{BaoZLZX2010,KangMPS2011,SchwartzAB2010,RoyPBMW2009}
\item [2/4] \cite{GodefroidKS2005,CadarDE2008}
            Historic: \cite{King1976,CadarE2005,CadarGPDE2006,SenMA2005}
\item [2/6] \cite{BrumleyCLNS2007,ChipounovKC2011}
            Optional: \cite{GaneshD2007,BucurUZC2011,BounimovaGM2012}
\item[2/11] \cite{GodefroidKL2008,MaPFH2011}
            Optional: \cite{BabicMMS2011,GodefroidL2011,ChaARB2012}
\item[2/13] \cite{Myers1999,JohnsonW2004}
\item[2/18] \cite{ChongLMQVZZ2007,MullerC2012}
\item[2/20] \cite{Malacaria2007,McCamantE2008}
            Historic: \cite{Denning1982}
\item[2/25] \cite{NewsomeMS2009,CastroCM2008}
            Optional: \cite{BackesKR2009,YasuokaT2010,PhanMTP2012}
\item[2/27] \cite{WahbeLAG1993,McCamantM2006}
            Historic: \cite{DeutschG1971}
\item [3/4] \cite{YeeSDCMOONF2009,SehrMBKPSYC2010}
\item [3/6] \cite{AnselMETCSSBY2011,MorrisettTTTG2012}
\item[3/11] \cite{AbadiBEL2005,DaviDEFHHNS2012}
            Optional: \cite{ZengTM2011,WangJ2010}
\item[3/13] \cite{WartellMHL2012,DhurjatiKA2006}
            Optional: \cite{NandaLLC2006,AkritidisCRCC2008}

\item[3/25] \cite{Shacham2007,MasonSMM2009}
\item[3/27] \cite{LinZX2010,LeeAB2011}

\item [4/3] \cite{GovindavajhalaA2003,ChanderMS2001}

\item [4/8] \cite{SturtonHWK2011,MolnarPSW2005}
            Optional: \cite{KoepfMO2012}
\item[4/10] \cite{MalkhiNPS2004,HuangEKM2011}
            Optional: \cite{SettyMBW2012,HuangKE2012}

\item[4/15] \cite{BarakGIRSVY2001,MadouASB2005}
\item[4/17] \cite{PachecoLEB2007,GaneshLR2009}

\item[4/22] \cite{JangAB2012,HackettDWY2006}
            Optional: \cite{BesseyBCCFHHKME2010,CifuentesKLHV2012}
\item[4/24] \cite{KleinEHACDEEKNSTW2009,UdreaLF2006}
            Optional: \cite{DietlDEMWCPP2010}

\item[4/29] \cite{EnckGCCJMS2010,YanY2012}
            Optional: \cite{EgeleKKV2011,EnckOMC2011,OcteauJM2012}
\item [5/1] \cite{Chlipala2010,PolitzEGK2011}
            Optional: \cite{WassermannS2007,ArtziKDTDPE2008,SamuelSS2011,BarthCS2009}

\item[e01] \cite{PachecoLEB2007,GaneshLR2009}
           Historic: \cite{MillerFS1990}
\item[e02] \cite{BrumleyS2004,HarrisJR2010}
           Optional: \cite{VaughanC2011}
\item[e03] \cite{GovindavajhalaA2003,SturtonHWK2011}
\item[e04] \cite{MolnarPSW2005,KoepfMO2012}
\item[e05] \cite{JangAB2012,HackettDWY2006}
           Optional: \cite{BesseyBCCFHHKME2010,CifuentesKLHV2012}
\item[e06] \cite{KleinEHACDEEKNSTW2009,UdreaLF2006}
           Optional: \cite{DietlDEMWCPP2010}
\item[e07] \cite{LinZX2010,LeeAB2011}
\item[e08] \cite{ReedP2010,BartheLPB2012}
           Optional: \cite{GaboardiHHNP2013}
\item[e09] \cite{MalkhiNPS2004,HuangEKM2011}
           Optional: \cite{SettyMBW2012,HuangKE2012}
\item[e10] \cite{BarakGIRSVY2001,MadouASB2005}
\item[e12] \cite{WassermannS2007,ArtziKDTDPE2008,Chlipala2010,PolitzEGK2011}
           Optional: \cite{SamuelSS2011,BarthCS2009}
\item[e13] \cite{EnckGCCJMS2010,EgeleKKV2011,EnckOMC2011,YanY2012}
           Optional: \cite{OcteauJM2012}

\end{itemize}

\bibliographystyle{alpha}
\bibliography{pas2013}

\end{document}
